## The Safer Nursing Care Tool as a guide to nurse staffing requirements on hospital wards: observational and modelling study

Peter Griffiths,<sup>1,2\*</sup> Christina Saville,<sup>1</sup> Jane E Ball,<sup>1,2</sup> Rosemary Chable,<sup>3</sup> Andrew Dimech,<sup>4</sup> Jeremy Jones,<sup>1</sup> Yvonne Jeffrey,<sup>5</sup> Natalie Pattison,<sup>4,6</sup> Alejandra Recio Saucedo,<sup>1</sup> Nicola Sinden<sup>7</sup> and Thomas Monks<sup>1,2</sup>

<sup>1</sup>School of Health Sciences, University of Southampton, Southampton, UK <sup>2</sup>National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care Wessex, University of Southampton, Southampton, UK <sup>3</sup>Training, Development & Workforce, University Hospital Southampton NHS Foundation Trust, Southampton, UK

<sup>4</sup>Clinical Services, The Royal Marsden NHS Foundation Trust, London, UK <sup>5</sup>Nursing & Patient Services, Poole Hospital NHS Foundation Trust, Poole, UK <sup>6</sup>School of Health and Social Work, University of Hertfordshire, Hatfield, UK <sup>7</sup>Nursing Directorate, Portsmouth Hospitals NHS Trust, Portsmouth, UK

\*Corresponding author Peter.griffiths@soton.ac.uk

**Declared competing interests of authors:** Peter Griffiths reports grants from the National Institute for Health Research (NIHR) during the conduct of the study [Health Services and Delivery Research (HSDR) 13/157/44, PR-ST-1115-10017 and HSDR 17/05/03)] and support from NIHR through a senior investigator award and as part of the NIHR Collaboration for Leadership in Applied Health Research (CLAHRC) Wessex. He is a member of the NHS Improvement safe staffing faculty steering group. The safe staffing faculty programme is intended to ensure that knowledge of the Safer Nursing Care Tool (SNCT), its development and its operational application is consistently applied across the NHS, using the evidence base underpinning the SNCT. This role attracts no remuneration. Jane E Ball reports grants from NIHR during the conduct of the study (PR-ST-1115-10017) and support from the NIHR CLAHRC Wessex. Rosemary Chable reports grants from NIHR during the conduct of the study (HSDR 13/157/44). Andrew Dimech reports that he was working on his clinical doctorate during the conduct of the study. Jeremy Jones reports grants from NIHR during the conduct of the study (II-LB-0814-20006, RP-PG-0610-10078 and HSDR 17/05/03). Thomas Monks reports grants from NIHR CLAHRC Wessex during the conduct of the study.

Published March 2020 DOI: 10.3310/hsdr08160

# **Plain English summary**

Safer Nursing Care Tool: observational and modelling study Health Services and Delivery Research 2020; Vol. 8: No. 16 DOI: 10.3310/hsdr08160

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

M any hospitals in England use the Safer Nursing Care Tool to help them decide the numbers of nurses and support workers to employ on wards. The nurse in charge assesses the patients who are on the ward and categorises them depending on how severely ill and dependent on nursing care they are. The tool provides nursing time estimates for each patient category and is used to decide how many nurses to employ. There is very little evidence about the use of this tool.

We used the tool to calculate the staff required every day in four hospital trusts. We looked to see if nurses were more likely to report that they had enough staff when they had the number estimated by the tool. After taking into account other factors that could affect this relationship, we found that when there are fewer staff than the tool suggests are needed, the chance of nurses reporting 'enough staff for quality' is lower. However, other factors not considered in the tool (ward type, single rooms, day of the week, time of day) also affect reported staffing adequacy.

The tool's guidelines suggest employing enough staff to meet demand on an average day. We compared this with higher, and lower, staffing levels using temporary staff to meet shortfalls. We developed a computer program that simulates the staff required, staff available and staff reassigned to other wards, in addition to the hiring of extra temporary staff. Employing fewer nursing staff can be cheaper but often there are not enough staff on the wards because temporary staff cannot make up the shortfall. This increases the risk of death and longer hospital stays for patients. Although it is more expensive, we found that employing more staff, at the level needed to meet the demand observed on 90% of days, could provide value for money because outcomes were improved at modest cost.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Griffiths et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

## **Health Services and Delivery Research**

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **HS&DR** programme

The HS&DR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HS&DR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/ health-services-and-delivery-research.htm

#### This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 14/194/21. The contractual start date was in May 2016. The final report began editorial review in May 2019 and was accepted for publication in October 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Griffiths *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### Editor-in-Chief of Health Services and Delivery Research and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk